Author's response to reviews

Title: Medical services of a multicultural summer camp event. Experiences from the 22nd World Scout Jamboree, Sweden 2011.

Authors:

Ib Jammer (ib.jammer@helse-bergen.no)
Christina Allansdotter Andersson (christinaallansdotter.andersson@skane.se)
Anna Lindholm Olinder (anna.lindholm.olinder@medsci.uu.se)
Bo Selander (bo.selander@skane.se)
Anna Elmerfeldt Wallinder (anna.elmerfeldt@gmail.com)
Stefan Rocco Hansson (Stefan.Hansson@med.lu.se)

Version: 4 Date: 11 April 2013

Author's response to reviews: see over
Dear Editor,

Thank you for the peer review, we uploaded the manuscript with some minor changes according to the remarks of the reviewer. Please see the reviewer comments (in italic) and our reply below.

1. **Page 4, first paragraph, last sentence: You need to add some references to support the statement on psychological issues.**
   
   We added a reference and rephrased the sentence to: Psychological issues at a camp can range from homesickness to neuropsychological disorders and sometimes behavioural or psychiatric episodes [11].

2. **Page 4, first paragraph, third line: consider replacing "self-destructiveness"; maybe with "self-harm behaviour".**
   
   We have rephrased the sentence, see comment 1.

3. **Page 4, healthcare at campsite section, third paragraph, second line: repetition of "jamboree Head Quarter". Consider replacing with "Jamboree's coordinator/etc".**
   
   We have replaced “Jamboree Head Quarter” in the second line. The sentence reads now: “The location was chosen in order to have a close proximity with core communication infrastructure and camp decision makers in case of a major crisis.”

4. **Page 4, healthcare at campsite section, fourth paragraph, second line: consider replacing "extremities". Do you mean "limbs" or something else such as "suspected fractures" or "extreme cases"?**
   
   We replaced “extremities” with “limbs” as we did X-ray of limbs for suspected fractures.

5. **Page 5, first sentence: multiple repetition of the word "contact". Consider alternative words.**
   
   We rephrased the sentences to:
   Of total 40,061 scouts, the medical service registered 3,229 (7.3%) visits. Of those 3,229 visits, 2,893 (89.6%) were primary contacts and 336 (10.6%) were follow-up appointments.

6. **Page 5, line starting with "128 x-ray": consider replacing "extremities" with "limbs".**
   
   We replaced “extremities” with “limbs”.

We are looking forward for your decision. Please contact us if you have further remarks regarding our manuscript.

Best regards,

Ib Jammer